

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gabardo, Francisco Adilson; Pereima, João Basilio; Einloft, Pedro

Article

The incorporation of structural change into growth theory: A historical appraisal

EconomiA

Provided in Cooperation with: The Brazilian Association of Postgraduate Programs in Economics (ANPEC), Rio de Janeiro

Suggested Citation: Gabardo, Francisco Adilson; Pereima, João Basilio; Einloft, Pedro (2017) : The incorporation of structural change into growth theory: A historical appraisal, EconomiA, ISSN 1517-7580, Elsevier, Amsterdam, Vol. 18, Iss. 3, pp. 392-410, https://doi.org/10.1016/j.econ.2017.05.003

This Version is available at: https://hdl.handle.net/10419/179657

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



NC ND https://creativecommons.org/licenses/by-nc-nd/4.0/



WWW.ECONSTOR.EU





Available online at www.sciencedirect.com



Economia

EconomiA 18 (2017) 392-410

www.elsevier.com/locate/econ

The incorporation of structural change into growth theory: A historical appraisal

Francisco Adilson Gabardo^{a,*}, João Basilio Pereima^{a,b}, Pedro Einloft^a

^a Federal University of Paraná, Brazil
 ^b The Graduate Program of Development Economics, Brazil
 Received 12 February 2016; received in revised form 23 March 2017; accepted 26 May 2017
 Available online 3 June 2017

Abstract

Despite being an empirical fact that structural change is an inseparable companion of the growth process, it appears as if growth theorists have relegated it to a secondary role. One of the reasons for this apparent neglect is undoubtedly the difficulty of dealing with the issues of sectoral dynamics and structural change within the framework of analytical models. A second reason derives from the fact that for a long time the analysis of growth, from a theoretical perspective, has focused predominantly on aspects of supply and technical progress, leaving the analysis of demand and consumption evolution, crucial for the understanding of structural change, aside. The present paper provides an overview of some of the main works in modern growth theory and appraises the introduction of the subject of structural change into the analysis of economic growth. The exposition elucidates the sources and effects of the process of structural change and surveys some of the recent literature from different schools of thought that integrates structural change into their analysis, commenting on their main features and contributions.

JEL classification: O100; O400

Keywords: Growth theory; Structural change; Kaldor facts; Kuznets facts; Dual economy

© 2017 The Authors. Production and hosting by Elsevier B.V. on behalf of National Association of Postgraduate Centers in Economics, ANPEC. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

1. Introduction

Although the process of economic growth might seem stable in the aggregate in the long-run, in historical perspective, the decline of the agricultural sector and the expansion of the industrial and the service sectors have led to a massive transformation of the economic landscape. The change in the relative importance of these three broadly defined sectors (agriculture, manufacturing, and services) that have accompanied the process of economic growth is one way to narrowly define structural change. However, structural change is a broader economic process that encompasses changes in the

* Corresponding author.

http://dx.doi.org/10.1016/j.econ.2017.05.003

E-mail addresses: franciscogabardo@ufpr.br (F.A. Gabardo), joaobasilio@ufpr.br (J.B. Pereima), pedroeinloft@gmail.com, einloft@ufpr.br (P. Einloft).

Peer review under responsibility of National Association of Postgraduate Centers in Economics, ANPEC.

^{1517-7580 © 2017} The Authors. Production and hosting by Elsevier B.V. on behalf of National Association of Postgraduate Centers in Economics, ANPEC. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

structure of production and employment within and between all sectors¹ of the economy as well as the emergence of new sectors and the disappearance of old ones.

The process of structural change have been documented by many authors. Fischer (1939), Clark (1940), Kuznets (1957) and Chenery (1960) are among the first ones to document it. Most of these early works were predominantly empirical without a deeper theoretical explanation regarding the sources and consequences of structural change. One early attempt to theorize the process of structural change was made by Clark (1940). In his work, Clark related the observed shifts in sectoral resources to two elements: differential productivity growth and Engel effects. By identifying these two elements, Clark laid down the basic theoretical relations that later would become the base of the theory of structural change (Syrquin, 1988, 213).

Perhaps the two early authors most cited for their work on structural change are Simon Kuznets and Luigi Pasinetti. Despite following different approaches,² these authors are praised for having emphasized the relevance of the role of structural change in the process of economic growth. Simon Kuznets,³ in particular, documented and analysed the process of structural transformation showing it to be an integral part of modern economic growth. His analysis gave rise to a broader search for uniform features or "stylized facts" of development and growth.

In his Nobel lecture, Kuznets summarized six characteristics of modern economic growth that emerged from his analysis based on conventional measures of national product and its components: population, labour force, and the like. The third of his six characteristics states that "the rate of structural transformation of the economy is high. Major aspects of structural change include the shift away from agriculture to nonagricultural pursuits and, recently, away from industry to services." (Kuznets, 1973, 248). Kongsamut et al. (2001) termed these empirical regularities the 'Kuznets facts' in analogy to the stylized facts established by Kaldor (1961). Luigi Pasinetti laid out his theory of structural change in two books published in 1981 and 1993: *Structural Change and Economic Growth* (1981) and *Structural Economic Dynamics* (1993). His theory will be analysed in more detail in Section 5.

In another important contribution, Maddison (1987) documented the same sort of regularity observed initially by Simon Kuznets. Maddison described the reallocation of labour in six major industrialized countries (France, Germany, Japan, Netherlands, U.K. and U.S.) showing that the average employment share in agriculture was 46.0% in 1870 and fell to 5.5% by 1984. During the same period, the average employment share in the service sector increased from 26.4% to 62.2%. The most commonly observed pattern of structural change is characterized by a systematic fall in the share of labour allocated to agriculture over time, by a steady increase in the share of labour in services, and by a hump-shaped pattern for the share of labour in manufacturing. Herrendorf et al. (2014) reported evidence of this pattern for currently rich countries.⁴

Despite appearing to be an obvious feature of empirical growth, structural change has not been fully integrated into the analysis of the growth process yet. There are some reasons that may explain why structural change has not being made part of growth theory. One of these reasons is the fact that many of the early models of growth were based on aggregate variables. While this approach highlights important aspects of the growth process, it is unable to analyse the driving forces of structural change. In order to incorporate structural dynamics into growth models, multi-sector⁵ models become necessary. Luigi Pasinetti recurrently pointed out the limitations of aggregate models. He has argued that,

Technical progress, productivity, consumption, investment, are no longer sufficient to define the economic system in a dynamic research. We must go beyond this, and find out what lies behind the façade of these aggregate expressions. In short, the research must be formulated in disaggregate terms. (Pasinetti and Spaventa, 1960, 1770).

A second reason is the fact that for a long time the analysis of growth has focused predominantly on aspects of supply and technical progress, leaving the analysis of demand and consumption evolution, crucial for the understanding

¹ In the literature, a sector is often equivalent to an industry. Thus, it is possible to say that the manufacturing sector is composed of several manufacturing industries.

 $^{^2}$ For more on the distinction of the two approaches refer to Syrquin (2012).

³ Kuznets (1966) documented the process of structural change for 13 OECD countries and the USSR between 1800 and 1960.

⁴ Belgium, Finland, France, Japan, Korea, Netherlands, Spain, Sweden, United Kingdom and United States.

⁵ The term 'multi-sector' refers here primarily to the assumption of multiple consumption sectors/goods.

of structural change, aside. This particular focus has led to the development of models ill-equipped to replicate the empirical features of structural change. Let us briefly elaborate on that.

Structural change is thought to originate from two sources. It is the result of sectoral differences in income elasticities of demand (demand side explanation), or of sectoral differences in productivity growth (supply side explanation). On the demand side, standard growth models normally assume homothetic preferences, which imply that all the goods have the same unitary income elasticities and that rich and poor individuals consume all the goods in the same proportion. On the supply side, they usually assume identical productivity growth across all sectors. Although not very realistic, these assumptions simplify the analysis and make models more tractable and suitable for analysing some aspects of growth, but not the process of structural change. According to Moshe Syrquin:

For Kuznets, and more generally in economic history and development, growth and structural change are strongly interrelated. Once we abandon the fictional world of homothetic preferences, neutral productivity growth with no systematic sectoral effects, perfect mobility, and markets that adjust instantaneously, structural change emerges as a central feature of the process of development and an essential element in accounting for the rate and pattern of growth. It can retard growth if its pace is too slow or its direction inefficient, but it can contribute to growth if it improves the allocation of resources by, for example, reducing the disparity in factor returns across sectors or facilitating the exploitation of economies of scale (Syrquin, 2012, 72).

Until the beginning of the 1990's most works on structural change were empirically-oriented contributions. Only recently that there has been a consistent effort to make structural change an integral part of modern growth theory. The present article provides an overview of some of the early, as well as more recent theoretical works in the literature of growth and assess the introduction of the subject of structural change into their analyses. Similar reviews in spirit were done by Krüger (2008) and Silva and Teixeira (2008). Krüger (2008) surveyed the existing research on structural change at various levels of aggregation with a special focus on the relation to productivity and technological change. He notes that, although in all the theories reviewed, technological progress drives structural change, it is the demand side which is crucial for determining which industries grow faster and which shrink, and therefore, shapes the direction of structural change (Krüger, 2008, 356). Silva and Teixeira (2008), following a different method, surveyed the literature on structural change through bibliometric analysis, combining citations from the area's 'seed journal' with a review of abstracts of all theoretical and empirical articles on structural change that were published over the past 40 years in the journals indexed in the Econlit. Silva and Teixeira's (2008) analysis reveals that most contributions put great emphasis on technology-driven growth and lack an appropriate treatment of the demand side.

This survey is organized as follows: the next section briefly presents some of the most influential works and models in growth theory since the end of World War II that, for different reasons, do not account for the process of structural change. Section 3 presents some early works on dual economy models, here considered the precursors of structural change growth models. Section 4 discusses the sources of structural change and presents some of the recent articles that have reconciled structural change and economic growth. Section 5 summarizes the Pasinetian model of structural change and growth using agent-based models. Section 7 presents some of the works that introduce structural change in the study of open economies. The review includes the New Latin American Structuralist (NLAS) growth theory, an approach that integrates increasing per capita income, sectoral dynamics, productive diversification and international specialization across countries. Finally, Section 8 presents our final remarks.

2. An economic growth theory without structural change

The exogenous and its successor the endogenous theory of economic growth can be understood as a long and arduous effort, which last almost half a century, dedicated to identify and isolate the determinants of sustained growth. Throughout this endeavour, the theory and its models focused on some specific determinants of economic growth such as saving and investment decision, human capital, learning by doing and R&D in an increasing returns to scale environment in order to explain sustained long -run economic growth. In some models, sustained growth is the result of a continuous innovation process generated by increases in labour productivity, capital productivity or both, in others it is generated by product diversification and creative destruction or by radical innovation. Sustained growth implies structural change and vice versa. However, structural change has been absent in modern growth theory, as described in more details in this section. It is interesting to note that structural change has been present in the business cycle

theory but not in growth theory. Despite the absence of the expression "structural change", there were some early attempts to explain macroeconomic fluctuations on the basis of industrial structural change made by Robertson (1915), Aftalion (1927), Frisch (1933)⁶ and Schumpeter (1939). Even Schumpeter referred to "industrial structure" rather than "structural change" and used the term in the same meaning as Alfred Marshall, namely industrial diversification, despite the known criticism from the first to the second. Modern growth theory chose to follow the aggregated approach, inspired by the nascent macroeconomic theory, either Keynesian or microfounded neoclassical.

2.1. One-sector exogenous growth models

After World War II, with a number of newly independent nations, the problem of promoting economic growth became a priority and growth theory came to occupy a central position in modern economics. The first models that tried to tackle the issue of economic growth were highly aggregated, and thus not intended for analysing the complexities of structural change. Among the most important articles published at that time are the works of Harrod (1939, 1948) and Domar (1946, 1947).⁷ Harrod's model extended the short run Keynesian static analysis into a long run dynamic one, and sought the answer to the question of what must be the rate of growth of income in a growing economy that equals investment and saving and guarantees a moving equilibrium through time. His analysis aimed at determining the necessary conditions to realize full-capacity utilization and full employment.

Harrod's article inspired a wave of contributions, including one of the most influential works in growth theory, Robert Solow's (1956) article *A Contribution to the Theory of Economic Growth*. Solow's original model was comprised of three equations: an aggregate production function that exhibits constant-returns-to-scale with smooth substitution and diminishing returns to capital and labour; an equation describing capital accumulation on the assumption of a constant rate of savings as a fraction of output; and a labour-supply function in which the labour force grows at an exogenous rate. Solow argued that an economy would automatically gravitate towards equilibrium if the relative price of labour and capital are flexible enough, and a spectrum of techniques exist so that the economy can move along a continuous production function combining different amounts of capital and labour.

In the Solovian model, the long-run equilibrium growth rate became independent of savings and investment decisions. Any increase in the economy saving or investment ratio would be offset by an increase in the capital–output ratio, leaving the long-run growth rate unchanged, as long as diminishing returns to capital exist. With the introduction of technical progress, diminishing returns to capital are offset, but then output and capital per worker grow at the same rate as technical progress, which grows at an exogenously given rate. Therefore, without technical progress the effects of diminishing returns would eventually cause economic growth to cease.

Throughout the 1960s, the original Solow model was extended into several directions. Solow himself modified his model introducing the notion of vintage capital with embodied technological progress and derived a new version without direct substitution between factors of production (Solow, 1959).

One particularly important extension was done by Cass (1965) and Koopmans (1965). They modified and microfounded the choice of consumption in the Solovian model using the intertemporal maximization developed by Ramsey (1928). The final model came to be known as the Cass–Koopmans–Ramsey model. The authors departed from the assumption of fixed saving rate and incorporated the permanent income and life-cycle savings hypotheses into the Solovian model. By incorporating these hypotheses, the saving behaviour becomes the result of explicit intertemporal utility maximization, i.e. the saving rate becomes endogenously determined. Therefore, unlike in the Solow model, where the saving rate is constant, in the Cass–Koopmans–Ramsey model, the saving rate may fluctuate along the transition to the long run steady state. The model also showed that, when exogenous technical progress is introduced, capital, consumption and output all grow at the same rate as the exogenous technical progress in the steady state.

Over the next three decades, mainstream growth theory focused predominantly on how to endogenize technical progress. Economists believe that technical progress is not exogenous, but actually depends on factors such as economic decisions made by profit seeking firms, funding of science and accumulation of human capital. If technical progress is an

⁶ A curiosity, Ragnar Frisch was who invented the term "econometrics" and was the first to use the words "microeconomics" to refer to the study of single firms and industries, and "macroeconomics" to refer to the study of the aggregate economy.

⁷ An important earlier contribution to growth theory is Young (1928).

endogenous variable then it has to be determined within the economic system. One of the problems with incorporating endogenous technical progress into growth theory is how to deal with the issue of increasing returns to scale.

2.2. One-sector endogenous growth models

One example of an early attempt to endogenize technical progress is Arrow (1962). Arrow used a mechanism called "learning by doing". The idea is that the growth of technical progress, commonly represented by the letter A is an unintended result of the experience of producing new capital goods. By assuming a process of learning by doing, Arrow was able to establish a link between investment in capital and technical progress.

Learning by doing was assumed to be external to both firms, the ones producing the new capital goods and the ones acquiring them, so they take the rate of technological progress as being given independently of their own production of capital goods. Thus, since in a competitive equilibrium no additional compensation would be paid to *A*, each firm maximizes profit by paying capital and labour their marginal products. Nevertheless, the growth of *A* becomes endogenous, in the sense that an increase in the saving propensity would affect its time path (Aghion and Howitt, 1998, 23). However, the model was fully worked out only in the case of a fixed capital-labour ratio and fixed (but vintage-specific) labour requirements. This implied that in the long run the growth of output was limited by growth in labour, thus in the absence of exogenous population growth, economic growth is no longer endogenous and becomes zero. In that sense, Arrow's model was not fully endogenous.

An important empirical regularity about the mechanism of learning by doing is that its external effect is bounded. So if the number of industries is fixed and their learning effects are bounded, growth cannot be sustained in the long run. A solution to this problem is to consider the possibility of new products being constantly introduced by way of learning externalities across industries and/or R&D, such mechanism was introduced in later models based on innovations. Despite its shortcomings, Arrow's idea of learning by doing formed the basis of the first type of endogenous growth models, which is known as the AK model.

The idea behind AK models is that growth is directly linked to two processes: learning by doing and capital accumulation. When firms accumulate capital, learning by doing generates technological progress, which is itself a kind of capital good (disembodied capital good), that tends to raise the marginal product of capital, thus offsetting the tendency for the marginal product to diminish when technology is unchanged. Hence, the source of increasing returns in AK models at the aggregate level is not capital accumulation itself, but the learning processes. According to this type of model, high growth rates are sustained by saving a large fraction of output, needed to finance a higher rate of technological progress.

The first AK model that accounted for sustained growth in per-capita output was that of Frankel (1962). Frankel wanted to reconcile the positive long-run growth result of Harrod–Domar with the factor-substitutability and marketclearing features of the Solovian model. In order to do that he used Arrow's learning by doing mechanism. He recognized that because individual firms contribute to the accumulation of technological knowledge when they accumulate capital, Harrod–Domar model did not require fixed coefficients and that aggregate productivity would depend upon the total amount of capital that has been accumulated by all firms. Therefore, as in Solow, Frankel's model displayed factor-substitutability (with Cobb–Douglas production technologies) and market-clearing, and as in Harrod, the model generated a long-run growth rate that depended on the saving rate, which was assumed to be constant.

Despite Frankel's earlier contribution, Romer (1986) and Lucas (1988) are considered the pioneering works of the neoclassical endogenous growth theory. Romer (1986) built an AK model, where productivity increased as a result of learning-by-doing externalities of the same sort as in Frankel (1962), but in which the constant saving rate was replaced by representative individual intertemporal utility maximization á la Ramsey.⁸

Lucas (1988), inspired by Becker (1964) theory of human capital, developed an AK model where the creation and transmission of knowledge occurs through human capital accumulation. He distinguishes two sources of human capital accumulation (or skill acquisition), namely education and learning by doing. By assuming that human capital accumulation involves constant returns to the existing stock of human capital, the model produces a positive growth rate in the steady state.

⁸ Ramsey (1928).

2.3. Multi-product endogenous growth models

A second wave of endogenous growth models came with the development of the so-called product-variety or expanding variety models. These are based on horizontal innovation, a type of innovation that causes productivity growth by creating new, but not necessarily improved, varieties of intermediate products. Horizontal innovation can be interpreted as a process innovation. Product-variety models do not deal directly with structural change, because they still have only one consumption or final good sector. However, since they focus on innovation and diversification, they contributed to the insertion of the issue on growth theory's agenda.

One of the advantages of the so-called product-variety endogenous growth models is the fact that they allow for a more explicit treatment of innovation and a better analysis of its structural effects. Most models based on horizontal innovation follow a similar structure consisting of three sequentially connected sectors: one sector that produces various 'designs' ('ideas', 'knowledge', etc.); a second sector that uses these 'designs' to produce various intermediate goods; and a third sector that uses the intermediate goods to produce the final good (and in most models, the final good is used-foregone-as an input in producing 'designs' and/or intermediate goods) (Park, 2010, 755). This differentiation between research, intermediate and final goods sectors add, to some extent, a notion of structural change into these models, but not the structural change in the sense of Kuznets.

Romer (1987), using the framework of monopolistic competition extended by Ethier (1982),⁹ developed an early version of a product-variety growth model. He assumes that productivity growth comes, not from learning externalities among individual firms, but from the continuous increase in the variety of specialized intermediate products, which prevents aggregate capital from running into decreasing returns. In this second model, Romer formalized an idea present in Young (1928), which is that growth can be sustained by the increased specialization of labour across an increasing variety of activities. As the economy grows, the larger market makes it worth paying the fixed costs of producing a large number of intermediate inputs, increasing the division of labour (specialization) and raising the productivity of labour and capital, which maintains growth (Aghion and Howitt, 1998, 36).

A more elaborated product-variety growth model was developed in Romer (1990), which is one of the most important contributions to the neoclassical endogenous growth theory. Romer extended his previous model and introduced a competitive research sector, which uses human capital and the existing stock of knowledge to generate new knowledge (blueprints or new designs) for new inputs (intermediate goods or machines) as a result of voluntary profit-motivated horizontal innovations. A monopolistically competitive intermediate-goods sector uses the new designs together with forgone output to produce a new variety of an intermediate good (new machines). The consumer goods sector produces final output using labour and intermediate goods, with a production function with a functional form borrowed from Ethier (1982). Final goods can be used for consumption and investment (in producing new blueprints).

In Romer (1990), there are two sources of increasing returns, namely specialization (increased labour division) and research spillovers. When a new design enables the production of a new intermediate good (new machine) there is an increase in the division (specialization) of labour, so an increase in the variety of machines, raises the productivity of labour. Additionally, a new design also increases the total stock of knowledge and thereby increases the productivity of human capital in the research sector, generating research spillovers. All researchers benefit from the accumulated knowledge *A* embodied in the existing designs. In other words, technological knowledge is a non-rival good. But at the same time knowledge is also excludable in the sense that intermediate firms must pay for the exclusive use of new designs which are monopolized by the firms which created them. In this model the aggregate production function exhibits constant returns to scale from the viewpoint of final good firms, which take the variety of machines as given, and increasing returns to scale for the economy as a whole.

The product-variety framework has been extended in several directions.¹⁰ Grossman and Helpman (1991, Chapter 4) model utility as a function of an expanding variety of consumer goods due to a sort of product innovation but where innovative products are in no way superior to older varieties, what eliminates the possibility of product obsolescence. In their framework real income increase because consumers show love-for-variety, meaning that the greater the number of differentiated varieties that the individual consumes, the higher his utility.

⁹ The basic approach to the benefits from variety comes from Spence (1976). Dixit and Stiglitz (1977) refined Spence's analysis and used to express consumer preferences over a variety of goods. Ethier (1982) applied this representation to inputs of production.

¹⁰ We refer the reader to Gancia and Zilibotti (2005) for a more thorough review.

Models based on horizontal innovation have contributed to the dissemination of an idea of structural change, since they acknowledge the existence of many heterogeneous capital goods in the real economy, which are produced by different sectors/technologies. Nevertheless, structural change in these models is limited to changes across capital-goods sectors, not the kind of broad sectoral dynamics described by the Kuznets facts.

The next stage in the evolution of growth models without structural change came with the emergence of vertical innovation models of quality improvement. This approach has come to be known as the Schumpeterian approach to endogenous growth and grew out of modern industrial organization theory. This type of models embody the force that Schumpeter (1942) called "creative destruction", and draw on the idea that growth is generated by a random sequence of quality-improving or also called vertical innovations that make existing products obsolete. In Schumpeterian models of endogenous growth, a new variety of intermediate good replaces the old one, and its use raises the technology parameter A.

Despite some earlier works,¹¹ the benchmark model of Schumpeterian endogenous growth was developed by Aghion and Howitt (1992), where there is a research sector which is portrayed as in the patent-race literature surveyed in Tirole (1988) and Reinganum (1989). In their model, growth results exclusively from technological progress generated by competition among research firms that create innovations. Each innovation consists of a new intermediate good, whose use as input allows more efficient methods to be used in producing the consumption good. Research firms are motivated by the prospect of monopoly rents that can be captured when a successful innovation is patented. The monopoly rents, however, will only last until the next innovation, which will render obsolete the existing intermediate good (Aghion and Howitt, 1992). While vertical and horizontal innovation based models complement each other, in both types of models, the old products disappear only through the introduction of new products, which means that the number of final products remains constant.

Much of the work developed by the exogenous and the early endogenous growth theory adopted an aggregated approach. Despite being insightful on the conditions required for a steady state growth path, the structure and diversification of economic system was neglected in favour of deepening the understanding of the determinants of growth and its effects. The later endogenous growth models, such as the product-variety and the Schumpeterian models have showed that an important mechanism of sustained growth is the rise of variety based on new and improved capital. The evolution of economic system through diversification implies a change in the aggregate productive structure. However, since in these models industries are treated as symmetric, they all expand evenly, making it impossible to explain major processes of structural changes, sector interdependence and transfer of resources across different industries. Even Aghion and Howitt (1998, 65) recognise that these models

[...] miss the stages of development in which resources are gradually reallocated from agriculture to manufacturing and then to services, all with different factor requirements and different technological dynamics. The economy is always a scaled up version of what it was years ago, and no matter how far it has developed already prospects for future development are always a scaled up version of what they were years ago.

In order to explain the major processes of structural changes, sector interdependence and transfer of resources across different industries models with more than one final good sector and where industries expand at different rates were necessary.

3. Dual economy: the precursor of growth and structural change theory

One-sector growth models have been extensively used by economists of all schools of thought. The main advantage of this type of model is their minimalist structure, what makes them highly tractable. One-sector models focus essentially on the growth process within a modern sector. However, they abstract from several features of the process of growth, making them not suitable for analysing the first stages of economic development and inter-sectoral phenomena such as industrialization or structural change.

By the beginning of the 1950s, some economists dissatisfied with the approach followed by aggregated growth theory, started to developed alternative frameworks better suited to describe the process of structural change that

¹¹ One of the earliest versions of a Schumpeterian endogenous growth model is Segerstrom et al. (1990). In their model, sustained growth arises from a succession of product improvements in a fixed number of sectors.

occurs in the first stages of economic development. One of these alternative approaches came to be known as dual economy model, and includes the works of Lewis (1954), Ranis and Fei (1961) and Jorgenson (1961) among others.

Dual economy models are commonly used to represent the first stages of an economy's development. A period in which, emphasis is laid on the balance between capital accumulation and population growth, each adjusting to the other. These models are based on the concept of structural heterogeneity, in which the economy is composed of two asymmetric sectors, a relatively advanced and a relatively backward sector. These asymmetries are not merely technological but also include institutional, behavioural, and informational aspects. These two sectors follow completely different economic logics, so they cannot be lumped together as the neoclassical theory does with the capital and the final goods sectors. They not only differ in terms of the goods produced, but also in terms of the nature of the growth process, wages and employment mechanisms.

In dual economy models, the traditional sector is characterized by subsistence wages, abundance of labour, low productivity, labour-intensive production process, no capital accumulation and no technical progress. In contrast, the modern sector is defined by higher wages as compared to the traditional sector, higher marginal productivity, capital-intensive production process, technical progress. Since dual economy models focus on inter-sectoral relationships and flows, growth in these models depend in large part on the rate at which resources, especially labour, can be transferred from the traditional to the modern sector. The two sectors have received other denominations in the literature such as capitalist and subsistence, modern and traditional, industry and agriculture, urban and rural and primary and secondary.

The first dual economy model was formalized by Lewis (1954). The Lewis model, as it came to be known, is a classical model consisted of an underdeveloped economy with two sectors: a traditional, overpopulated subsistence sector characterized by zero marginal labour productivity and a high-productivity modern sector. There is an unlimited supply of labour available at subsistence wage, so labour is able to move to the modern sector without lowering output. Moreover, capitalists save everything, but workers (and landlords) save nothing. The unlimited supply of labour from the traditional sector keeps wages from increasing in the modern sector and ensures that capital accumulation in that sector is sustained over time. Both labour transfer and the modern sector employment growth are brought about by output expansion in the modern sector. The speed with which this expansion occurs is determined by the rate of investment and capital accumulation in the modern sector. Thus, the source of structural transformation can be found in the unlimited supply of labour from the traditional sector. The Lewis model underlines the importance of transfers of resources from low-productivity to high-productivity activities in the process of economic development.

Jorgenson (1961) is another seminal example of a dual economy model. He develops a neoclassical version of Lewis dual economy model. Jorgenson's model assumes two sectors: agriculture and industry. Agricultural output depends on labour and on a fixed amount of land. Industrial output depends on labour and on capital. The production functions have the Cobb–Douglas form, with constant returns to scale and neutral technological change. Only labourers in the advanced sector can be assumed to respond to wage differentials between employment opportunities in agriculture and industry. Industrial wage-rate is equal to the marginal product of labour.

In Jorgenson's model the division of labour between the two sectors is straightforward: if there is no agricultural surplus, all labour remains in the agricultural sector; if an agricultural surplus can be generated, labour is released from the land and transferred to the manufacturing sector at a rate that is equal to the rate of growth of the agricultural surplus. Manufacturing production is only possible if some initial capital stock exist, however small it may be, the model shows that there is no critical level of initial capital endowment below which no sustained growth is possible. Even the smallest initial stock can give rise to sustained growth. Once the initial injection of capital is made, capital accumulation in the manufacturing sector continues at a pace determined by the growth of the labour force in that sector and by the terms of trade between the two sectors. The modern sector is also subject to technical progress, so the more rapid the rate of growth in the advanced sector. Eventually, the economy is dominated by the development of the advanced sector and becomes more and more like the advanced economic systems described neoclassical model and less like a dual economy.

In dual economy models, the assumption of unlimited supply of labour is crucial, because it allows the possibility of having increasing saving and in turn capital accumulation and faster growth without increasing wages. However, this process does not continue forever. Eventually, the withdrawal of labour from the traditional sector reaches the point at which the marginal product of the remaining labour rises to equality with the subsistence wage. From this point on, wages rise in both sectors as growth continues and workers can choose and offer their labour to the highest bidder (the

traditional or the modern sector). A labour market, in a neoclassical way, is formed. The dualistic structure is over and the economy enters a new stage of development characterized by scarcity of labour.

There is another early class of models that also divided the economy into two sectors: a capital sector and a consumption goods sector. These models are different from dual economy models. Despite having two sectors they were not intended at analysing structural transformation. In this class of model all the final good sectors (agriculture, manufacturing, service) have the same structure, thus being aggregated into one unified "consumption" goods sector. Examples of this type of models are Meade (1961), Uzawa (1961), Kurz (1963) and Takayama (1963, 1965).

Dual economy models can be seen as one of the first attempts to model economic growth taking into account a process of structural change. They provided important insights on the determinants and on the outcomes of sectoral structural dynamics. However, the models were limited in scope and not fully equipped to incorporate all the driving mechanisms of structural change. More comprehensive models were required in order to fully analyse the complexities involved in the process of structural transformation. Some of these more comprehensive models are discussed in the next section.

4. Structural change and growth

Recently, a new wave of growth models have developed better ways of analysing the process of structural change and some have being able to reconcile structural change with the broad patterns of aggregate balanced growth. The literature have identified several channels to explain the process of structural change, ranging from demand-driven factors to purely technological determinants. The mechanism adopted to combine structural change and growth depend in part on the assumptions made about the origin of the structural change and in part on the characteristics of the model, if it is balanced or non-balanced.

A first explanation of the sources of structural change relies on differences in income elasticities of demand across sectors. This explanation has been dubbed 'utility-based' and suggests that if one assumes non-unitary expenditure elasticities of demand, then increases in real per-capita expenditure levels affect the sectoral expenditure shares leading to the reallocation of labour across sectors. In order to capture these differences in income elasticity of demand across sectors, non-homothetic preferences have to be assumed. Authors that locate the origin of structural change in long-run changes in consumer tastes have incorporated structural dynamics into their models through the assumption of non-homothetic utility functions. These are multi-sector models consistent with Engel's law and in which structural change is driven by differences in the income elasticity of demand across goods. Engel's law states that as a household's income increases, the fraction that it spends on food (agricultural products) declines. Several authors have found that Engel's law holds not only for food, but it is a more general law of consumption.¹² According to Houthakker (1987) Engel's law is one of the most robust empirical findings in economics.

A second explanation is based on changes in relative prices. These changes affect the expenditure structure whenever the elasticity of substitution across sectors is different from one. This view of structural change is known in the literature as the 'technological' explanation. Authors who adopt this view incorporate structural change into their models through two mechanisms: (1) by assuming differential productivity growth across sectors or; (2) by assuming changes in the relative prices of inputs if sectors vary in the intensity with which they use inputs and there are changes in the relative supply of factors (Herrendorf et al., 2014). Despite their differences, these two mechanisms can account for structural change from a supply side perspective. Some authors see the utility-based and the technological explanations as competing theories. However, there is not enough evidence in the literature that proofs that they cannot coexist. Some authors were able to derive structural change from both sources simultaneously, Boppart (2014) is one example.

4.1. Structural change and non-balanced aggregate growth

Another characteristic of models that integrate structural change and growth is related to the issue of reconciling structural change with the Kaldor facts. The Kaldor stylized facts propose that the growth rate, the interest rate, the capital output ratio, and the labour share are roughly constant over time while capital per worker and real wage grows

¹² Deaton and Muellbauer (1980) concluded that the vast majority of studies obtains the result that the expenditure share of a product changes systematically with income.

11. 64. 1

401

over time (Kaldor, 1961). These regularities provide a good characterization of the long run behaviour of the U.S. economy and are believed by many to be a reasonable approximation of the long-run growth experience of a modern economy. Consistency with the Kaldor facts is not a feature present in all the models that combine growth with structural change. The reason for this might be found in the interpretation of the Kaldor facts. According to Pasinetti, Kaldor never interpreted his 'facts' as an empirical justification for the construction of a theory of balanced growth. Nicholas Kaldor himself did not claim that any of the regularities he had uncovered would be constant at all times. Some authors, for different reasons, decided to focus on structural change and abstract from the empirical regularities of the Kaldor facts. Some of these works are summarized in this section.

Baumol (1967) is one of the first articles to model structural change based on the technological explanation in which unbalanced growth is a general feature of the growth process because different sectors grow at different rates owing to different rates of technological progress. He built a model consisted of two sectors, a 'progressive' and a 'non-progressive' one and of a single factor, labour. In the progressive sector, as a result of continuing technological progress, labour productivity grows cumulatively at a constant compounded rate r, and in the non-progressive sector labour productivity is constant. Baumol's models makes four propositions: (1) through time the non-progressive sector unitary cost of output, relative to that of the progressive sector, will rise without limit; (2) unless demand for it is highly inelastic, output in the non-progressive sector will decline and perhaps approach zero; (3) in order to maintain balanced growth, represented by a constant ratio between the outputs of the two sectors, the share of the labour force allocated to the non-progressive sector must approach unity; (4) an attempt to maintain balanced growth, in a world of unbalanced productivity, will lead to a zero rate of growth of real output-per-capita.

Matsuyama (1992) assessed the role of agriculture productivity in the process of industrialization. He builds a twosector model of endogenous growth, where he is able to combine Engel's Law with learning-by-doing externalities in the industrial sector. The labour market is competitive and the wage rate is equalized across the two sectors. He makes two key assumptions in his article: (a) preferences are non-homothetic and the income elasticity of demand for the agricultural good is less than unitary, and (b) manufacturing productivity rises over time because of learning-bydoing. Agricultural productivity is determined purely exogenously. The model also analyses the relationship between industrialization and the assumptions concerning the openness of the economy. For the closed economy case, the model finds that an exogenous increase in agricultural productivity shifts labour from agriculture to manufacturing and thereby accelerates economic growth. However, for the open economy case, there exists a negative link between agricultural productivity and economic growth. An economy with less productive agricultural sector allocates more labour to manufacturing and will grow faster, in contrast, an economy with a more productive agricultural sector squeezes out the manufacturing sector de-industrializing over time and growing slower. Some limitations of the model are the fact that agricultural productivity is determined purely exogenously, there is no learning-by-doing in agriculture and technological advances in manufacturing do not improve agricultural productivity.

More recently, Echevarria (1997) introduced structural change on a dynamic general equilibrium model. Her paper explains the relation between income levels and rates of growth as an effect of changes in sectoral composition driven by different income elasticities for primaries, manufacturing and services. She assumed three different consumption goods (primary, manufacturing and services) demanded by agents displaying non-homothetic preferences. There are two factors of production: labour and capital, and each consumption good is produced using different factor intensities. Capital is produced in the second sector (manufacturing) and distributed among all the sectors. Since the productivity rate in each sectors is different and exogenously determined, the growth rate of the economy is affected by changes in sectoral composition, which is in turn driven by non-homothetic preferences (Echevarria, 1997, 431). The assumption of nonhomotheticity of preferences is crucial and drives the results in the model. Echevarria explains that a poor country, which consumes mainly necessities (primary), cannot save (invest) much. As it gets richer, it will invest or save more, thereby encouraging growth. At the same time production will shift to the second sector (manufacturing), which has a higher rate of technical change; thus, the first effect is reinforced. Both effects, increase in investment and increase in average total factor productivity, imply an acceleration in the growth rate. Yet, the savings rate (net savings or investment as percentage of GDP) does not increase monotonically over time, eventually the savings rate falls, thus driving the growth rate down. If, at the same time, production shifts to the third sector (service) with its lower rate of technical change, that will reinforce the reduction in the growth rate (Echevarria, 1997, 445).

Echevarria's model predicts the following regularities: (1) a hump-shaped correlation between growth rates and income levels, with poor countries having the lowest rate of growth and middle-income countries having the highest growth rate; (2) the higher proportion of agriculture on GDP in poor countries, and the higher proportion of services

on GDP in rich countries; (3) the comparatively more expensive services in rich countries; (4) the larger share of the labour force employed in agriculture in less-developed countries, and the larger share employed in services in developed countries; and (5) the higher share of output paid to labour in rich countries. Echevarria concludes that sectoral composition explains an important part of the variation in growth rates observed across countries.

Following a different approach, Park (1998) analyses structural change focusing on structural transformation from an agriculture-based to a manufacturing-based economy. He developed a three-factor, three-goods endogenous growth model with a non-homothetic Stone–Geary type utility function to analyse transitional dynamics of structural change. The three factors are land, unskilled labour and the capital stock, which includes physical and human capital. The three goods are agricultural goods, manufacturing goods and new capital goods. In Park's model the long-run growth rate of the economy is determined by the size of the capital-producing sector, but the growth rate of each good is different on the balanced growth path as well as during the transitional period and although the capital-producing sector is assumed to be the engine of economic growth, the sectoral growth rate of this sector is not necessarily highest. Stone–Geary utility is useful and convenient, since it allows the presence of subsistence level and the possibility of income elasticity of demand for goods be less than unitary. This kind of utility function implies that there is a minimum, or subsistence level of food consumption that the household must consume. After this level has been achieved, the household starts to demand other items, in this case manufactured goods. Therefore, Stone–Geary utility has the ability to yield different growth rates during the transitional period. Park shows that sectoral contributions of respective industries to economic growth are variable and different and that structural change in production and factor use favours the manufacturing industry as opposed to the agricultural industry during the transitional period.

Laitner (2000) shows that structural change can affect the economy's saving rate through the operation of Engel's law taking account the composition of assets in household portfolios. Assuming the existence of non-homothetic preferences, Laitner builds a model with two sectors: agriculture and manufacturing. The model is based on the following structure: household saving follows the stages of life-cycle behaviour with overlapping generations, where each household lives for two periods, and is identical to all others born at the same time. There are no inheritances or bequests and the household takes prices as given. Young households will save all labour earnings and retired households will spend all their wealth. This pattern do not change over time even if incomes change, but the composition of consumption depends on changes in income. On the production side, aggregate effective labour supply depends on the number of young households and current technology. Exogenous technological progress raises per-capita income over time. In Laitner's model, a household whose standard of living is low cares only about agricultural consumption, but a household with a high standard of living, on the other hand, becomes satiated with agricultural products and devotes its remaining expenditures exclusively to manufactured goods. Hence, Engel's law implies a demand shift from agriculture to manufacturing goods as income rises. Consequently, the economy goes from an initial position where it specializes in agriculture to devoting more and more labour to manufacturing production. In the limit, the share of agriculture in total GDP tends to zero and the share of manufacturing converges to unity.

Another work where structural change is explained through differences in income elasticities of demand across sectors applying non-homothetic preferences of Stone–Geary type is Caselli and Coleman (2001). They made a joint study of the U.S. structural change (the decline of agriculture as the dominating sector) and regional average wage convergence. The authors attempted to explain the decline of agriculture as the dominating sector in the United States and the convergence of income per-capita across the U.S. States. They used human capital accumulation to explain discrepancies in labour and output trends in the decline of agriculture. The model features a closed economy with two locations, North and South; two goods, farm and manufacturing; and three factors of production, land, labour, and capital. They assumed that North and South are equally good at producing manufactures but the South enjoys a comparative advantage in the production of farm goods. The non-homotheticity is represented by a less than unit income elasticity of demand for farm good.

Gollin et al. (2002) built a model of structural change to explain why the process of industrialization occurs at different dates and why it proceeds slowly. The authors, like others, also make use of a non-homothetic utility function of the Stone–Geary variety in order to generate structural transformation. The model uses a basic neoclassical framework modified to include both an agricultural and a non-agricultural sector. Countries begin the process of industrialization only after they satisfy their basic agricultural needs. Hence, low agricultural productivity can substantially delay industrialization. Asymptotically, agriculture's employment share shrinks to zero, and the model becomes identical

to the standard one-sector neoclassical growth model. One important conclusion is that improvements in agricultural productivity can accelerate the start of industrialization and so, have large effects on a country's relative income.

4.2. Structural change and balanced aggregate growth

Although balanced aggregate growth is not an inevitable property of growth models, most of them strive to be consistent with the Kaldor facts. One of the properties of balanced aggregate growth models is that the fraction of capital and labour allocated to different industries remain constant over time. Hence, *prima facie*, combining structural change at the sectoral level with the Kaldor facts at the aggregate level seems to be non-trivial. However, some authors were able to explain a transition along which, aggregate variables exhibit an almost balanced growth path, while there is sectoral change. Some of the works that have reconciled structural change and the Kaldor facts are summarized in this section.

Kongsamut et al. (2001) is one of the first articles that have reconciled structural change and the Kaldor facts. In their article, the authors build a three-sector model displaying what they defined as a generalized balanced growth path, which is a trajectory consistent with the dynamics of structural change and along which the real interest rate is constant. Their approach is also based on Engel's Law, with structural change being driven by income effects. The authors used a Stone–Geary utility function, which Acemoglu describes as being "[...] a highly tractable way of introducing income elasticities that are different from one for different subcomponents of consumption and Engel's Law" (Acemoglu, 2009, 699). They directly manipulated the utility function in order to generate different income elasticities of the goods and sectors. Income elasticity of demand was assumed less than one for agricultural goods, equal to one for manufacturing goods, and greater than one for services. Although being perhaps the most popular in the literature, especially in models with two goods,¹³ this Stone–Geary-type specification has some disadvantages. It is only applicable when working with a small number of goods. Moreover, in order to be consistent with the generalized balanced growth path, Kongsamut et al. (2001) had to rely on a widely criticized knife-edge condition, which ties together preference and technology parameters and implies constant relative prices.

In another article, Ngai and Pissarides (2007) formalized the first mechanism that derives structural change purely from different sectoral total factor productivity (TFP) growth rates. They show that, given a low (below one) elasticity of substitution between the final goods produced by each sector, and assuming that all goods have unit income elasticity, different TFP growth rates predict sectoral employment changes. Ngai and Pissarides' model contains many consumption goods and a single capital good, supplied by a manufacturing sector. They assume identical Cobb–Douglas production functions in all sectors except for their rates of total factor productivity growth. Each sector produces a differentiated good that enters a constant elasticity of substitution (CES) utility function. The model is able to predict a shift of employment away from sectors with high rate of technological progress toward sectors with low growth, and eventually, in the limit, all employment converges to only two sectors, the sector producing capital goods and the sector with the lowest rate of productivity growth (Ngai and Pissarides, 2007, 438). In order to satisfy Kaldor's stylized facts of aggregate growth a logarithmic intertemporal utility function is required. Ngai and Pissarides's (2007) results are consistent with the evidence concerning the decline of agriculture's employment share, the rise and then fall of the manufacturing share, and the rise in the service share. The key requirement for their results is a low substitutability between final goods. Ngai and Pissarides's model confirms Baumol's (1967) claim that the production costs and prices of the stagnant sector should rise indefinitely and labour should move in the direction of the stagnant sector. However, it contradicts Baumol's conclusion that as more weight is shifted to the stagnant sector, the economy's growth rate will be on a declining trend and eventually converge to zero. The reason for the contrasting result is that Ngai and Pissarides included capital in their analysis, which was left out in Baumol (1967).

Foellmi and Zweimüeller (2008) developed a growth model consistent with both the Kaldor and Kuznets facts, based on the assumption of hierarchical preferences. Their model shows that reallocation of labour is driven by differences in income elasticities across sectors. The basic idea of their analysis is that households expand consumption along a hierarchy of needs, in which goods are weighted according to their essentiality. In order to depict the equilibrium process of growth and structural change consistent with the Kaldor facts, the 'hierarchy function', which characterizes the willingness of consumers to move from high priority to low priority goods, must take a particular form with some

¹³ See e.g. Matsuyama (1992).

specific characteristics. In their model, the authors adopt a particular form of power function. Foellmi and Zweimüeller's model is capable of generating movements not only of labour out of agriculture and into services, but also a hump shape in the evolution of the manufacturing share, with a period of increasing manufacturing employment followed by a period of de-industrialization, replicating empirically observed patterns.

Differently from other models that also adopt the utility-based explanation for structural change with non-homothetic preferences, Foellmi and Zweimüeller' model introduces a situation where new goods are continuously introduced. Each new goods starts out as a luxury with a high income elasticity and ends up as a necessity with a low income elasticity. According to the authors, these non-linearities in Engel curves generate consumption cycles that account for structural change. For the sake of simplicity and to highlight the demand channel, the authors assume exogenous and identical sectoral productivity growth across all sector, the relative price structure remains constant over time. Despite addressing the issue of structural change along a growth path consistent with the Kaldor facts Foellmi and Zweimüeller's model relies on income effects alone in order to explain structural change.

Acemoglu and Guerrieri (2008) construct a model that allows for differences in capital intensities across sectors. In their model, capital deepening and sectoral factor intensity differences are the determinants of the relative price dynamic. The authors build a two-sector general equilibrium model with constant elasticity of substitution preferences and Cobb–Douglas production technologies, where they show that, capital deepening increases the relative output of the more capital-intensive sector while simultaneously induces a reallocation of capital and labour away from that sector (Acemoglu and Guerrieri, 2008). They show that, provided the elasticity of substitution is less than one, one of the sectors (typically the more capital-intensive one) grows faster than the rest of the economy, but given that the relative prices move against this sector, its (price-weighted) value grows at a slower rate than the rest of the economy. Moreover, the model shows that capital and labour are continuously reallocated away from the more rapidly growing sector, thus generating sectoral structural change. Convergence to equilibrium is slow with the capital share in national income and the interest rate varying only by relatively small amounts.

Despite reconciling structural change with balanced growth, in Acemoglu and Guerrieri's model the Kaldor facts hold only asymptotically. Moreover, regarding the demand side the authors do recognize the importance of income effects but abstract from non-homotheticity of preferences, placing the source of structural change solely on the supply side of the economy.

It is possible that the two explanations for structural change described above, the utility-based and the technological explanation are not mutually excludable, but actually complementary. Boppart (2014) provides empirical evidence that both drivers of structural change are relevant. Moreover, he builds a model integrating both explanations. He combines non-homothetic preferences and differential TFP growth while reconciling the Kaldor facts with structural change. Boppart relies on non-Gorman preferences where the marginal propensity to consume a "good" and a "service" differs between rich and poor households and inequality affects the aggregate demand structure. The author conducts a structural estimation that allows for the decomposition of the structural change into an income and a substitution effect showing that both channels of structural change are of roughly equal importance. The empirical analysis shows that the model's functional form fits the data and the framework can replicate the observed structural change quantitatively (Boppart, 2014). While advancing in the understanding of the interplay between the two drivers of structural change, Boppart's model do not contemplate the possibility of creation of new sectors.

The role of the Kaldor facts in describing the path of growth of modern capitalist economies of the XXI century is, at least, debatable. For instance, Jones and Romer (2010) points out that the Kaldor facts might be outdated since they revolved around a single state variable, namely physical capital. They argue that the facts should be updated and include variables such as ideas, institutions, population, and human capital. Another article that questions the validity of the Kaldor facts is Jorgenson and Timmer (2011). The authors examine whether Kaldor's stylized facts provide an accurate description of more recent structural changes. The authors especially question Kaldor's stability of the share of labour in GDP over time, they find that the labour share in value-added is declining and that the decline is pervasive in all sectors and regions, except in US finance and business services. How well the Kaldor facts really represent long-run economic growth is an open question. Especially regarding developing and less developed countries.

Some authors believe that general equilibrium and aggregate balanced models may not be the best way to represent long-run economic growth. In the next sections we summarize two schools of thought that follow a different methodology to analyse the issue of structural change, the Passinetian model of structural change and the agent-based evolutionary growth theory. These schools do not see the economy as a balanced system or as an unbalanced one heading towards equilibrium; they see the economic system as being in constant change, in constant evolution. For these schools disequilibrium is the normal state of the economy.

5. Pasinetti's structural change

An important contribution to the development of the theoretical foundations of structural change was given by Luigi Pasinetti. One of the merits of his contribution was to make the issue of structural change the centre of his analysis of growth. Pasinetti shifted the focus of the analysis from the determinants of growth to the structural dynamics of the economic system. His theory of structural dynamics was laid out in two books published in 1981 and 1993. The name of his books are illustrative of his interest in explaining de origin and effects of structural change on growth, and especially, on the development of societies: *Structural Change and Economic Growth* (1981) and *Structural Economic Dynamics* (1993).

Pasinetti follows a completely different approach from the authors surveyed in the previous sections. His theory is based on post-Keynesian and classical elements and stresses the inevitability of structural change. While for some authors structural change is a transitional step towards balanced growth, in Pasinetti's approach disequilibrium and instability are the normal state of affairs. Economic growth is seen as a process of continuous change. Pasinetti's view of structural change is based on what he calls a pure production model. This model abstracts from institutions of the economic system and the behavioural modes of economic agents and concentrates on the natural or primary characteristics that enable a system to grow.

The prime mover of structural change in Pasinetti's analysis is technical progress as a result of exogenous learning activity. Technical progress in one particular sector has two effects. The first effect is the reduction of the labour coefficient in that sector and consequently an increase in productivity. Increases in productivity lead to increases in per-capita income. These two things are inseparable as Pasinetti puts it: "Increases in productivity and increases in income are two facets of the same phenomenon. Since the first implies the second, and the composition of the second determines the relevance of the first, the one cannot be considered if the other is ignored." (Pasinetti, 1981, 69). Increases in per-capita income "endows single individuals with the possibility of obtaining larger amounts, or a larger number, of goods and services, or entirely new goods and services altogether." (Pasinetti, 1993, 37). However, consumers do not expand their demand for all goods proportionally, consumption expansion follows Engel's law, when consumption of a good has become satiated, only then attention turns to the next higher good in the hierarchical ordering. The second effect of technical progress is the emergence of new products. Thus, over time, the sector's labour and demand coefficients are modified by technical progress and by changes in consumer's tastes. If the rate of variation of these two coefficients are equal, the economy expands the various sectors proportionately, and its structure remains unchanged over time. This is what happens in the traditional models of exogenous growth. However, nothing guarantees that these rates will be the same. If the rates are different, as normally happens, the economy experiences structural dynamics of employment. Hence, in Pasinetti's analysis, structural dynamics of technology and demand generates structural dynamics of employment.

One of the weakness of Pasinetti's model was the assumption that the labour and demand coefficients were exogenously determined. This weakness was overcome in different ways by different authors. Reati (1998) and Fusari and Reati (2013) introduced long waves in the Pasinettian model in order to endogenize technical change. Andersen (2001) developed an evolutionary micro founded model based on a set of rules that made endogenous the demand coefficients, the labour coefficients, and the number of available sectors. Araujo (2013) merged the Pasinettian model with the Kaldorian literature to endogenize technical change by using sectoral Kaldor–Verdoorn laws.

6. Agent-based evolutionary growth theory

The last generation of endogenous growth introduced the mechanism of creation-destruction as an explanation for long-run growth. Following a similar vein, the evolutionary theory of growth introduced a mechanism of productive diversification that can be interpreted as a type of structural change. With its roots in the works of Nelson and Winter (1982) and Dosi (1982) among others, the evolutionary growth theory's early works focused on the issues of changes in technology and innovation, concentrating less on full analytic solutions and more on illustrative simulations including agent-based modelling. Models that specifically relate growth with structural change are fairly recent in this literature.

One example of such model is Montobbio (2002). The author analyses the determinants of structural change and aggregate productivity growth based on the aggregation of the behaviours of heterogeneous firms in different economic sectors. Using an evolutionary approach, Montobbio sheds new light on the determinants of sectoral shifts and nonuniform growth. The main purpose of his analysis is to show that productivity growth, at sectoral and aggregate levels, is the result of a general evolutionary process. In order to do that the model explores the properties that connect the distribution of heterogeneous firms to the growth of industries, and the mechanisms that account for changes in the relative weight of different sectors (i.e. structural change) within an economy. Aggregate productivity growth is analysed as the result of the aggregation and interdependence of heterogeneous firms and sectors and is guided by the selection mechanisms within and between industries. Moreover, he analyses the impact of sector-specific income elasticities of demand. One of Montobbio's findings is that the aggregate growth rate of labour productivity depends negatively upon the covariance between the sectoral elasticities of demand and the sectoral average unit costs, positively on the exogenous rate of demand growth and is proportional to the variance between sectoral unit costs and to the average of firms' unit cost variances within sectors.

Following a different approach, Saviotti and Pyka (2004) developed an evolutionary model where growth and structural change are driven by the creation of new sectors. The authors constructed a dynamic model of growth that involves qualitative change, where economic development is driven by the creation of new industrial sectors. In their model, they put forward two hypotheses that link variety to economic development: (1) the growth in variety is a necessary requirement for long-term economic development, and (2) variety growth, leading to the development of new sectors, and productivity growth in pre-existing sectors, are complementary and not independent aspects of economic development. The authors justify these two hypotheses based on the imbalance between productivity growth and demand growth raised in Pasinetti (1981, 1993). According to Saviotti and Pyka, if the economy were constituted by a constant set of activities, in presence of growing productivity it would become possible to produce all demanded goods and services with a decreasing proportion of the resources used as inputs, including labour. This imbalance would then constitute a bottleneck for economic development and could lead to technological unemployment. The introduction of new goods and services can be a way of compensating for the potential displacement of labour and of other resources.

Another article that tackles the issue of structural change from an evolutionary perspective is Ciarli et al. (2010). The authors offer a theoretical analysis of long-run economic growth as an outcome of structural changes. In an agent-based micro-founded framework they investigate the properties of a growth model that embeds the relation between technological and organizational change, income distribution and the dynamics of consumption affecting macroeconomic growth. Microeconomic behaviours are modelled in line with the large and consolidated evolutionary theory of technical change and economic growth, while the macro-framework draws from the structuralist literature. They observe and explain the interactions between technological change, firm organization, income distribution, consumption behaviour and growth. Ciarli et al. confirm the relevance and interdependence of these structural changes and underline their microeconomic sources.

Agent-based evolutionary models are in some ways more powerful in that they facilitate the modelling of complex interactive processes. Agent-based evolutionary models present many advantages over more traditional approaches, including the possibility of considering simultaneously demand-side factors and technological progress, the emergence of new goods and sectors and the feedback effects between structural change and aggregate growth, a task not so trivial to accomplish within an analytical model. However, the greatest advantage of this type of models is the possibility of bringing together elements that are disconnected and isolated in the literature within a coherent theoretical framework.

7. Structural change and international trade

In recent years, there has been an effort to develop an integrated theory that includes increasing per capita income, sectoral dynamics and productive diversification. Structural change theory has focused in explaining the growth process of a country as a closed economic system. Scarce attention has been paid to the effects of structural change in an open economy, which may lead to international specialization across countries having significant impacts on world growth inequality.

Different approaches have been developed to extend the analysis of structural change to open economies. One of these approaches is based on an extension of Thirlwall's Law,¹⁴ which Araujo and Lima (2007) called Multi-Sectoral Thirlwall's Law (MSTL). This approach explores the connection between a country's sectoral composition in its trade and the differences in income elasticities of demand across sectors. Aggregate income elasticities are calculated as weighted averages of the income elasticities of exports and imports from each sector, where the weights are the sector's shares in exports and imports, respectively (Romero and McCombie, 2016). According to Araujo and Lima (2007) even if the sectoral elasticities and the growth rate of world income are constant, it is possible to raise a country's long-term growth rate by favourably changing its sectoral composition of the its trade. This conclusion illustrates that, structural change may not be solely a by-product of the process of economic growth but an important factor capable of influencing it.

In a more recent work, Araujo (2013) merged the Pasinettian model with the Kaldorian literature and endogenized technical change combining Thirlwall's Law and Verdoorn's Law¹⁵ in a multi-sectoral cumulative causation growth model. The model takes into account supply and demand factors simultaneously along with their related effects on economic growth. Technological progress, induced by exogenous foreign demand, increases productivity through static and dynamic increasing returns to scale. Higher productivity results in higher real per capita income that are in turn converted into higher per capita demand, which further increases productivity. This virtuous cycle renders structural changes endogenous. The author extended the result obtained in Araujo and Lima (2007) in order to highlight the influence of technological progress upon the balance of payments (BoP) constrained growth rate. Moreover, Araujo (2013) not only showed the possibility of reconciling cumulative causation with the view of BoP constrained, but also that cumulative causation is one of the mechanisms that explains the ever-widening per capita and technological gaps amongst rich and poor nations. One of the conclusions of the model is that a region that produces a final good with high income elasticity of demand will have high technological progress and thus exhibit faster growth rates in general, than a region that produces a final good with low elasticity of demand (Araujo, 2013, 138).

A slightly different approach was developed by what can be called the New Latin American Structuralist (NLAS) growth theory. This approach combines three theoretical branches: the Latin American Structuralism (LAS), originally due to Prebisch (1950), ECLAC (1954), Cimoli (1988) and more recently to Cimoli and Porcile (2010, 2014); growth theory led by demand under balance of payment constrained (Thirlwall, 1979) and technological change and its effects on international trade (Dosi et al., 1990). NLAS is grounded on the empirical observation that countries which export goods associated with higher productivity levels grow more rapidly, even after controlling for initial income per head, human capital levels, and time-invariant country characteristics. This empirical regularity is described in Hausmann et al. (2006).

According to the NLAS approach, the level of diversification and technological development of a country's productive structure conditions its insertion in the international trade which, in turn, determines its balance of payment (BoP) equilibrium and in the end, its national gross product consistent with these BoP equilibrium. A country with low productive diversification, due to low technological capabilities, tend to specialize in the production of few commodities in order to explore the relative, but restrict advantages. Since this economy do not produce many of the final and intermediate goods that it requires, its dependency on imports is high. The income elasticity of the goods this economy imports tend to be high, while the income elasticity of its exports tend to be low. This combination produces a permanent restriction on growth. If the rate of growth remains high for consecutive periods, an external crisis emerges to correct the BoP deficit, reducing the growth rate. The external crisis can be postponed by attracting capital inflow through high interest rates. However, high interest rates discourage investment leading to a situation of insufficient demand. Growth in this economy cannot be sustained, after some periods accumulating external deficits the system exhibit its internal contradiction and growth without fail stops. The way out of this situation would be to increase the share of technology-intensive sectors in its economic structures.

The NLAS approach falls in line with the Schumpeterian and endogenous theory of growth when it locates the source of long-run sustained growth in the innovation and diversification processes. Yet, the NLAS approach differs

¹⁴ Thirlwall's Law states that each country's equilibrium growth rate must correspond to the ratio between its income elasticity of demand for exports and its income elasticity of demand for imports, multiplied by the growth rate of external demand (or world income) (Romero and McCombie, 2016).

¹⁵ In general terms, Verdoorn's law implies the existence of a stable and positive relationship from the growth rate of manufacturing output to the growth rate of labour productivity in manufacturing.

from the traditional one, in two aspects. The first is the recognition of the important role played by aggregated demand, which becomes the main channel through which the restrictions operate. The second is the integration of structural change and growth into the international trade literature.

One example of the NLAS literature is Cimoli et al. (2010). In their model the authors depart from the Schumpeterian assumption that countries which increase the share of technology-intensive sectors in their economic structures benefit more from technological learning and innovation. These countries would be better equipped to respond to changes in international markets and to compete in sectors in which demand grows at higher rates. The authors find evidence that certain kinds of structural change (namely those in which technologically intensive sectors increase their participation in the economy) favour growth. Structural change is measured by the Krugman Index and by the participation of high-tech and medium-tech exports in total exports. Both variables show a positive association with relative rates of growth in the international economy.

In the NLAS approach, structural change is embedded in the idea of productive diversification, which is captured by an aggregate index that represents the number of different goods the economy is able to produce and export. Structural change is analysed from a macroeconomic standpoint and, therefore, different from the evolutionary micro-approach where structural change emerges from the decision of individual firms. Despite being a macroeconomic approach, the NLAS has the advantage of allowing for the analysis of the insertion of a country or region into the international economy.

8. Final remarks

Modern growth theory has gone through a series of transformations since its birth. The focus of its analysis has expanded and the features of the models have become more complex. What we have seen in recent years is a new wave of models that are incorporating structural change into growth theory in ways never attempted before. These models have contributed to the understanding of the mechanisms behind the relationship between growth and structural change and have pushed further the frontier of the field of growth theory. However, as Acemoglu points out "[...] we are still far from a satisfactory framework for understanding the process of sectoral reallocation of factors [...]" (Acemoglu, 2009, 720). Moreover, the majority of the models that analyse structural change focus on the reallocation of productive factors from some sectors of the economy to others, normally from agriculture to manufacturing and then to services. However, Jorgenson and Timmer (2011) argues that the classical trichotomy among agriculture, manufacturing, and services may have lost most of its relevance. They have discovered enormous heterogeneity among different services subsectors, largely ignored in the previous literature, something that calls for greater attention to individual service sectors to understand the process of economic growth and structural change. Thus, the next step in the evolution of modelling structural change has to account for sector heterogeneity. In addition, the majority of growth models that deal with structural change assume the number of sectors and/or products in the economy to be constant. From the analytical standpoint, this simplification makes models more tractable. However, if one wants to understand the interactions between sectoral dynamics and growth, models should account for the increase/decrease in the number of sector/products in the economy. Lastly, the analysis of the impacts of structural change on international trade and its feedback effects on growth must also be considered if a comprehensive theory of growth and structural change is to be developed.

References

Acemoglu, D., 2009. Introduction to Modern Economic Growth. Princeton University Press, Princeton.
Acemoglu, D., Guerrieri, V., 2008. Capital deepening and nonbalanced economic growth. J. Polit. Econ. 116, 467–498.
Aftalion, A., 1927. The theory of economic cycles based on the capitalistic technique of production. Rev. Econ. Stat. 9, 165–170.
Aghion, P., Howitt, P., 1992. A model of growth through creative destruction. Econometrica 60, 137–147.
Aghion, P., Howitt, P., 1998. Endogenous Growth Theory. The MIT Press, Cambridge, Massachusetts.
Andersen, E.S., 2001. Satiation in an evolutionary model of structural economic dynamics. J. Evol. Econ. 11, 143–164.
Araujo, R.A., 2013. Cumulative causation in a structural economic dynamic approach to economic growth and uneven development. Struct. Change Econ. Dynam. 24, 130–140 http://www.sciencedirect.com/science/article/pii/S0954349X1200063X.

Araujo, R.A., Lima, G.T., 2007. A structural economic dynamics approach to balance-of-payments-constrained growth. Camb. J. Econ. 31, 755–774. Arrow, K.J., 1962. The economics implications of learning by doing. Rev. Econ. Stud. 29, 155–173.

Baumol, W.J., 1967. Macroeconomics of unbalanced growth: an anatomy of urban crisis. Am. Econ. Rev. 55, 297-307.

Becker, G.S., 1964. Human Capital, 1st ed. Columbia University Press, New York.

- Boppart, T., 2014. Structural change and the Kaldor facts in a growth model with relative price effects and non-Gorman preferences. Econometrica 82, 2167–2196.
- Caselli, F., Coleman, J., 2001. The U.S. structural transformation and regional convergence: a reinterpretation. J. Polit. Econ. 109, 584-616.

Cass, D., 1965. Optimum growth in an aggregative model of capital accumulation. Rev. Econ. Stud. 32, 233-240.

- Chenery, H.B., 1960. Patterns of industrial growth. Am. Econ. Rev. 50, 624-654.
- Ciarli, T., Lorentz, A., Savona, M.M.V., 2010. The effect of consumption and production structure on growth and distribution: a micro to macro model. Metroeconomica 61, 180–218.
- Cimoli, M., 1988. Technological gaps and institutional asymmetries in a north-south model with a continuum of goods. Metroeconomica 39, 245–274, http://dx.doi.org/10.1111/j.1467-999X.1988.tb00878.x.
- Cimoli, M., Pereira, W., Porcile, G., Scatolin, F., 2010. Structural change, technology, and economic growth: Brazil and the CIBS in a comparative perspective. Econ. Change Restruct. 44, 25–47, http://dx.doi.org/10.1007/s10644-010-9095-3.
- Cimoli, M., Porcile, G., 2010. Specialization, wage bargaining and technology in a multigoods growth model. Metroeconomica 61, 219–238, http://dx.doi.org/10.1111/j.1467-999X.2009.04066.x.
- Cimoli, M., Porcile, G., 2014. Technology, structural change and BOP-constrained growth: a structuralist toolbox. Camb. J. Econ. 38, 215–237, http://dx.doi.org/10.1093/cje/bet020.

Clark, C., 1940. The Conditions of Economic Progress. McMillan & Co., London.

- Deaton, A., Muellbauer, J., 1980. Economics and Consumer Behavior. Cambridge University Press, Cambridge, USA, 24th printing 2009.
- Dixit, A.K., Stiglitz, J., 1977. Monopolistic competition and optimum product diversity. Am. Econ. Rev. 67, 297–308.
- Domar, E.D., 1946. Capital expansion, rate of growth, and employment. Econometrica 14, 137–147.
- Domar, E.D., 1947. Expansion and employment. Am. Econ. Rev. 37, 34-55.
- Dosi, G., 1982. Technological paradigms and technological trajectories: a suggested interpretation of the determinants and directions of technical change. Res. Policy 11, 147–162.
- Dosi, G., Pavitt, K., Soete, L., 1990. The Economics of Technical Change and International Trade. New York University Press, Washington Square, New York.
- Echevarria, C., 1997. Changes in sectoral composition associated with economic growth. Int. Econ. Rev. 38, 431-452.
- ECLAC, 1954. Estudio económico de América Latina. Economic Commission to Latin American and Caribbean ECLAC, United Nation.

Ethier, W.J., 1982. National and international returns to scale in the modern theory of international trade. Am. Econ. Rev. 72, 389-405.

- Fischer, A., 1939. Primary, secondary and tertiary production. Econ. Rec. 15, 24-38.
- Foellmi, R., Zweimüeller, J., 2008. Structural change, Engel's consumption cycles and Kaldor's facts of economic growth. J. Monet. Econ. 55, 1317–1328.
- Frankel, M., 1962. The production function in allocation and growth: a synthesis. Am. Econ. Rev. 52, 995–1022.
- Frisch, R., 1933. Propagation problems and impulse problems in dynamic economics. In: The Economics of Structural Change vol. II, Critical Writings (2003). Edward Elgar, Aldershot.
- Fusari, A., Reati, A., 2013. Endogenizing technical change: Uncertainty, profits, entrepreneurship. A long-term view of sectoral dynamics. Struct. Change Econ. Dynam. 24, 76–100 http://www.sciencedirect.com/science/article/pii/S0954349X12000409.
- Gancia, G., Zilibotti, F., 2005. Horizontal innovation in the theory of growth and development. In: Aghion, P., Durlauf, S. (Eds.), Handbook of Economic Growth, vol. 1. Elsevier, pp. 111–170 (Chapter 3).
- Gollin, D., Parente, S., Rogerson, R., 2002. The role of agriculture in development. Am. Econ. Rev. 92, 160-164.
- Grossman, G.M., Helpman, E., 1991. Innovation and Growth in the Global Economy. MIT Press, Cambridge.
- Harrod, R.F., 1939. An essay in dynamic theory. Econ. J. 49 (193), 14-33.
- Harrod, R.F., 1948. Towards a Dynamic Economics. MacMillan and Co., London.
- Hausmann, R., Hwang, J., Rodrik, D., 2006. What you export matters. J. Econ. Growth 12, 1-25, http://dx.doi.org/10.1007/s10887-006-9009-4.
- Herrendorf, B., Rogerson, R., Valentinyi, k., 2014. Growth and structural transformation. In: Aghion, P., Durlauf, S. (Eds.), Handbook of Economic Growth, vol. 2B, 1st ed. North Holland, Amsterdam, New York, pp. 855–941 (Chapter 6).
- Houthakker, H.S., 1987. Engel curve. In: Eatwell, J., Milgate, M., Newman, P. (Eds.), The New Palgrave Dictionary of Economics. Macmillan, London, pp. 142–143.
- Jones, C.I., Romer, P.M., 2010. The new Kaldor facts: ideas, institutions, population, and human capital. Am. Econ. J. Macroecon. 2, 224–245, http://dx.doi.org/10.1257/mac.2.1.224.
- Jorgenson, D.W., 1961. The development of a dual economy. Econ. J. 71, 309–334, http://dx.doi.org/10.2307/2228770.
- Jorgenson, D.W., Timmer, M.P., 2011. Structural change in advanced nations: a new set of stylised facts. Scand. J. Econ. 113, 1–29, http://dx.doi.org/10.1111/j.1467-9442.2010.01637.x.
- Kaldor, N., 1961. Capital accumulation and economic growth. In: Lutz, F.A., Hague, D.C. (Eds.), The Theory of Capital. St. Martin's Press, New York, pp. 177–222.
- Kongsamut, P., Rebelo, S., Xie, D., 2001. Beyond balanced growth. Rev. Econ. Stud. 68, 869–882, http://dx.doi.org/10.1111/1467-937X.00193.

Koopmans, T.C., 1965. On the concept of optimal economic growth. In: The Econometric Approach of Development Planning. Elsevier, North Holland, Amsterdan.

- Krüger, J.J., 2008. Productivity and structural change: a review of the literature. J. Econ. Surv. 22, 330–363, http://dx.doi.org/10.1111/j.1467-6419.2007.00539.x.
- Kurz, M., 1963. A two-sector extension of swan's model of economic growth: the case of no technical change. Int. Econ. Rev. 4, 68–79.

Kuznets, S., 1957. Quantitative aspects of the economic growth of nations: II. Industrial distribution of national product and labor force. Econ. Dev. Cult. Change 5, 1–111.

Kuznets, S., 1966. Modern Economic Growth. Yale University Press, New Haven, CT.

- Kuznets, S., 1973. Modern economic growth: findings and reflections. Am. Econ. Rev. 63, 247-258 http://www.jstor.org/stable/1914358.
- Laitner, J., 2000. Structural change and economic growth. Rev. Econ. Stud. 67, 545–561.
- Lewis, W.A., 1954. Economic development with unlimited supplies of labour. Manch. School 22, 139–191, http://dx.doi.org/10.1111/j.1467-9957.1954.tb00021.x.
- Lucas, R.L., 1988. On the mechanism of economic development. J. Monet. Econ. 22, 3-42.
- Maddison, A., 1987. Growth and slowdown in advanced capitalist economies: techniques of quantitative assessment. J. Econ. Lit. 25, 649-698.
- Matsuyama, K., 1992. Agricultural productivity, comparative advantage, and economic growth. J. Econ. Theory 58, 317–334.
- Meade, J.E., 1961. A Neo-classical Theory of Economic Growth. Oxford University Press, New York.
- Montobbio, F., 2002. An evolutionary model of industrial growth and structural change. Struct. Change Econ. Dynam. 13, 387-414 http://www.sciencedirect.com/science/article/pii/S0954349X02000061.
- Nelson, R.R., Winter, S., 1982. An Evolutionary Theory of Economic Change. Harvard University Press, EUA.
- Ngai, L.R., Pissarides, C.A., 2007. Structural change in a multisector model of growth. Am. Econ. Rev. 97, 429-443.
- Park, M.S., 2010. Capital and interest in horizontal innovation models. Camb. J. Econ. 34, 755–772, http://dx.doi.org/10.1093/cje/bep074.
- Park, S., 1998. Transitional dynamics of structural changes. Seoul J. Econ. 11, 75–99.
- Pasinetti, L., Spaventa, L., 1960. Verso il superamento della modellistica aggregata nella teoria dello sviluppo economico. Riv. Polit. Econ. 15, 1751–1781.
- Pasinetti, L.L., 1981. Structural Change and Economic Growth. Cambridge University Press, Cambridge.
- Pasinetti, L.L., 1993. Structural Economic Dynamic: A Theory of the Economic Consequences of Human Learning. Cambridge University Press, Cambridge.
- Prebisch, R., 1950. The economic development of Latin America and its principal problems. Econ. Bull. Lat. Am. 7.
- Ramsey, F.P., 1928. A mathematical theory of saving. Econ. J. 38, 543-559.
- Ranis, G., Fei, J.C.H., 1961. A theory of economic development. Am. Econ. Rev. 51, 533-565 http://www.jstor.org/stable/1812785.
- Reati, A., 1998. A long-wave pattern for output and employment in Pasinetti's model of structural change. Econ. Appl. 51, 27–75.
- Reinganum, J., 1989. The timing of innovation: research, development, and diffusion. In: R, S, R, W (Eds.), Handbook of Industrial Organization, vol. 1, 1st ed. Elsevier, pp. 849–908 (Chapter 14), URL: http://EconPapers.repec.org/RePEc:eee:indchp:1-14.
- Robertson, D.H., 1915. A Study of Industrial Fluctuation: An Enquiry into the Character and Causes of the So-called Cyclical Movements of Trade. P.S. King and Son, London.
- Romer, P.M., 1986. Increasing return and the long run growth. J. Polit. Econ. 94, 1002–1037.
- Romer, P.M., 1987. Growth based on increasing returns due the specialization. Am. Econ. Rev. 77, 56-62.
- Romer, P.M., 1990. Endogenous technological change. J. Polit. Econ. 98 (Pt. II), S71-S102.
- Romero, J.P., McCombie, J.S.L., 2016. The multi-sectoral thirlwall's law: evidence from 14 developed European countries using product-level data. Int. Rev. Appl. Econ. 30, 301–325, http://dx.doi.org/10.1080/02692171.2015.1102207.
- Saviotti, P.P., Pyka, A., 2004. Economic development by the creation of new sectors. J. Evol. Econ. 14, 1–35.
- Schumpeter, J.A., 1939. Business Cycles: A Theoretical, Historical, and Statistical Analysis of the Capitalist Process. McGraw-Hill, New York, London.
- Schumpeter, J.A., 1942. Capitalism, Socialism and Democracy. Harper and Brothers, New York.
- Segerstrom, P.S., Anant, T.C.A., Dinopoulos, E., 1990. A Schumpeterian model of the product life cycle. Am. Econ. Rev. 80, 1077–1091.
- Silva, E.G., Teixeira, A.A.C., 2008. Surveying structural change: seminal contributions and a bibliometric account. Struct. Change Econ. Dynam. 19, 273–300 http://www.sciencedirect.com/science/article/pii/S0954349X08000192.
- Solow, R.M., 1956. A contribution to the theory of growth. Q. J. Econ. 70, 65-94.
- Solow, R.M., 1959. Investment and technical progress. In: Mathematical Methods in Social Sciences. Stanford University Press, Stanford, CA, pp. 89–104, 1960.
- Spence, M., 1976. Product selection, fixed costs, and monopolistic competition. Rev. Econ. Stud. 43, 217–235.
- Syrquin, M., 1988. Patterns of structural change. In: Chenery, H., Srinivasan, T. (Eds.), Handbook of Development Economics, vol. 1. North Holland, Amsterdam, pp. 203–273 (Chapter 7).
- Syrquin, M., 2012. Two approaches to the study of structural change and economic development: Kuznets and Pasinetti. In: Arena, R., Porta, P.L. (Eds.), Structural Dynamics and Economic Growth, 1st ed. Cambridge University Press, Cambridge, pp. 69–87.
- Takayama, A., 1963. On a two-sector model of economic growth: a comparative statics analysis. Rev. Econ. Stud. 30, 95–104 http://www.jstor.org/stable/2295807.
- Takayama, A., 1965. On a two-sector model of economic growth with technological progress: a comparative statics analysis. Rev. Econ. Stud. 32, 251–262 http://www.jstor.org/stable/2295830.
- Thirlwall, A.P., 1979. The balance of payments constraint as an explanation of international growth rate differences. BNL Q. Rev. 32, 45–53 https://ideas.repec.org/a/psl/bnlaqr/197901.html.
- Tirole, J., 1988. The Theory of Industrial Organisation. The MIT Press, Cambridge, Massachusetts.
- Uzawa, H., 1961. On a two-sector model of economic growth. Rev. Econ. Stud. 29, 40-47 http://www.jstor.org/stable/2296180.
- Young, A.A., 1928. Increase returns and economic progress. Econ. J. 38, 527-542.